Author's response to reviews

Title: Is No News Good News? Inconclusive genetic test results in BRCA1 and BRCA2 from patients and professionals’ perspectives.

Authors:

Audrey Ardern-Jones Mrs (audreyaj2@gmail.com)
Regina (joint first author) Kenen Prof (kenen@tcnj.educ)
Elly Lynch Ms (elly.lychn@austin.org.au)
Rebecca Doherty Ms (becdoherty@yahoo.com.au)
Rosalind Eeles Dr (rosalind.eeles@icr.ac.uk)

Version: 4 Date: 25 November 2009

Author’s response to reviews: see over
Dear Editors,

I have included the title page in the manuscript and it includes the Title, Author list, Affiliations, E-Mail addresses and corresponding author. All these are the same as on the manuscript submission system.

I have included the Abstract and it is identical in the submission system.

I have placed the Authors' contributions section towards the end of the manuscript text, directly preceding the Acknowledgements. All authors read and approved the final manuscript.

I have included the Competing interests' section between the Conclusions and Authors' contributions and have written 'The authors declare that they have no competing interests'.

I have made the minor revisions.
I filled in all the references that included the term 'et al'

With kind regards,

Audrey Ardern-Jones